Math 144 Fall 2006 R. Schultz

GENERAL INFORMATION — SEPTERMBER 29, 2006

INSTRUCTOR: R. Schultz OFFICE: Surge 221 HOURS: 10:30–11:20 Mondays and by appointment TELEPHONE: (951) 827-6459 [if no answer, email messages are better] EMAIL: schultz@math.ucr.edu, reinhard.schultz@ucr.edu TEXTS: Halmos, Naive Set Theory; Lipschutz, Schaum’s Outlines: Set Theory ... ONLINE COURSE DIRECTORY: math.ucr.edu/∼res/math144 The latter has lecture notes, course outlines, general course information, homework assignments and a few other items. There will be a log of changes to the directory in the text file

aaaLOGFILE

to indicate what has been added or changed.